Author's response to reviews

Title: Systemic chemotherapy with doxorubicin, cisplatin and capecitabine for metastatic hepatocellular carcinoma

Authors:

Se Hoon Park (hematoma@gilhospital.com)
So Young Kwon (sykwonmd@gilhospital.com)
Oh Sang Kwon (kos@gilhospital.com)
Sun Suk Kim (ssk@gilhospital.com)
Ju Hyun Kim (jhkim@gilhospital.com)
Yeon Ho Park (pyh@gilhospital.com)
Jeong Nam Lee (jinlee@gilhospital.com)
Soo-Mee Bang (smbang@gilhospital.com)
Eun Kyung Cho (ekcho@gilhospital.com)
Dong Bok Shin (dbs@gilhospital.com)
Jae Hoon Lee (jhlee@gilhospital.co)
Yuna Lee (yuna712@medimail.co.kr)
Sang Hoon Han (btfulo@empal.com)

Version: 4 Date: 20 December 2005

Author's response to reviews:

Dear Editor

BMC CANCER

Title: Phase Systemic chemotherapy with doxorubicin, cisplatin and capecitabine for metastatic hepatocellular carcinoma

Please find attached revised manuscript and revision details. We really thank for the thoughtful review on our manuscript. Thank you for your acceptance of this manuscript.

Sincerely yours,

Se Hoon Park, M. D. / Dong Bok Shin, M. D.

#1. Please address direct if the primary tumor was included for tumor assessment.
we specified directly as "Primary tumor was included for response assessment."

#2. Methods - Please document what type of consent was sought from the study participants.
We specified that all patients had to provide a signed written informed consent before study inclusion.

#3. Tables must be divided into cells/fields.
We made tables with Microsoft Word format (cells/fields)

#4. Figures - The figures are cropped as closely as possible to minimize white space around the image.
we cropped figure 1 as you recommended.

#5. References - The reference list should contain all authors' names. The term "et al" should not be used.
Endnote output style (default for BMC Cancer) made "et al". Anyway, we edited by hand as your comment.